Author’s response to reviews

Title: The Cameroon Mobile Phone SMS (CAMPS) Trial: the Protocol for a Randomized Controlled Trial of Mobile phone text messaging versus usual care for improving adherence to HAART

Authors:

Lawrence CE Mbuagbaw (mbuagbawl@yahoo.com)
Lehana Thabane (thabanl@mcmaster.ca)
Pierre Ongolo-Zogo (pc.ongolo@gmail.com)
Richard T Lester (rlester.id@gmail.com)
Edward Mills (emills@cfenet.ubc.ca)
Jimmy Volmink (jvolmink@sun.ac.za)
David Yondo (dyondo@yahoo.fr)
Marie-José Essi (essimj@yahoo.com)
Renée-Cecile Bonono-Momnougui (bonono2003@yahoo.fr)
Robert Mba (robertmariemba@yahoo.fr)
Jean Serge Ndongo (njserge@yahoo.fr)
Francis Nkoa (fcnkoa@yahoo.com)
Henri Atangana Ondoa (atanganaondoa@yahoo.fr)

Version: 3 Date: 29 November 2010

Author’s response to reviews:

Dear Editor,

This is the protocol for a randomized clinical trial investigating the use of mobile phone text messages to improve adherence to antiretroviral medication. Sub-optimal adherence is one of the main reasons for treatment failure and the development of resistant viral strains. The World Health Organisation has recognised the potential of new technologies in improving health outcomes in resource limited settings. We are investigating the possible benefits of the Short Message Service with a particular motivational content. This publication will serve as a source of clarification for researchers interested in the trial results while presenting specific methodological features.

Formatting changes have been made, ethics and funding documentation have been provided.

Kind regards.